Author's response to reviews

Title: The development and pilot testing of a rapid assessment tool to improve local public health system capacity in Australia

Authors:

Prue Bagley (pruebagley@bigpond.com)
Vivian Lin (v.lin@latrobe.edu.au)

Version: 4 Date: 4 October 2009

Author's response to reviews:

We have made additions to the references where requested.
We have amended the in-text citations that were incorrect - changing from [7 p.3] to ([7] p.3) etc.
The tables have been added to the manuscript.
The abstract in the submission system matches the abstract in the manuscript.